 
CENWP-OP-B                                                                                                                                   2 July 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 26 of 2001, which covers the period 24 through 30 June.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 112.5 to 134.4kcfs.  Daily spill averaged 0.0 kcfs.  Daily project tailwater elevation averaged 11.0’ to 12.3’ msl.  Daily project forebay elevation averaged 73.6’ to 74.6’ msl.  River temperature averaged 63 to 65° F.  

1.2 Spill bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, remain open 4” during daylight hours.    

1.3 Unit 3 remained out of service for turbine rehab.  Unit 17 remains out of service for its four-year overhaul.  Unit 5 remains out of service for turbine rehab.  Unit 15 FGE summer tests began 15 June.  The unit is on  standby each night following FGE testing.  Unit 2 went out of service at 0454 on 25 June until 1420 on 28 June for disconnect work.  

1.4 As reported last week, Unit 15 went out of service on 23 June because an STS became stuck in the B gatewell.  FGE testing was suspended.  The unit returned to service at 0910 on 26 June after divers helped remove the STS.  FGE tests resumed that night.

1.5 Powerhouse Two fish units remain in service.  They provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

2.  ACTIVITIES. 
2.1 AWS Closures.   FV4-3 and FV4-4 at B-branch and FV5-3 and FV5-4 are all operating in manual until the new computer system is installed.  FV3-9 was closed for cleaning from 1541 until 1551 on 26 June and 1527 until 1545 on 28 June.

2.2 STS/VBS Inspections.  Five units were video inspected this reporting week.

Unit
Date
Run Hours
Comments

11
6/27
663
Good

12
6/27
641
Good

13
6/27
568
Good

14
6/27
464
Good

18
6/27
673
Good

Table 1.  STS/VBS Inspection Data.


2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage this reporting week.

3.  RESEARCH.

3.1 WDFW Fish Counting.  WDFW continues 16-hour visual counts and 8-hour video counts at Bradford Island and Washington Shore ladders.

3.2 NMFS Saltwater rearing. The Saltwater rearing facility is in operation. 

3.3 OCFWRU Barging condition study.  Began 24 April.

3.4 PSMFC Adult PIT.  Looking at A and B branch and Cascade Island for noise interference.  Analyzing data collected at the Washington Shore ladder.

3.5 PSMFC Smolt Monitoring Program.  Collection continues 24h/d at the monitoring facility.  24 h/d flat plate monitoring at DSM1.

3.6 WES Hydroacoustic Monitoring.  Monitoring began 20 April.

3.7 UI Barging condition study. They began sampling juvenile fish from transport barges on 22 April.

3.8 USGS Juvenile Radiotracking.  Radio tagging began 1 April.  Tracking is in progress as well.

3.9 USGS Adult EMG tracking.  Completed all tracking 30 June.  

3.10 UI Adult Fallback. UI Adult Radiotracking/Tagging.  PCC entrance gates at Powerhouse One are being operated according to UI request until 31 October.  Tagging of Chinook began at the AFC&MF 2 April and will continue seven days a week until the end of September. 

3.11 CRITFC/ODFW/WDFW.  Tagging Spring Chinook and steelhead on Tuesdays and Thursdays.

3.12 WDFW.  Tagging Spring Chinook with jaw tags in conjunction with their tooth-net studies.

3.13 NMFS FGE/OPE.  Work began 23 April.  Spring testing ended ~ 23 May.  Summer tests began 11 June.  

3.14 NMFS Lamprey.  Collection of Lamprey began June 1.  Spillway ladder entrances manually operated at night for a .5’ differential.  NMFS is holding extra lamprey for the Warm Springs Tribe, and collecting fish for USGS.

3.15 Pelfrey Sturgeon Hatchery.  Collecting broodstock in PH1 tailrace. 
4.  FISHWAYS. 
4.1 Project biologists inspected every day (24-30 June) this reporting week.  

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE
TR/WAR

All week
PH2 collection channel electronic velocity meter.
PH2 is not yet working.
Not fully installed.
New velocity meters installed in channel, awaiting core drilling at PH2.  


All week
FG3-28
Out of service
Manually closed because of displaced diffuser valve cover.
New covers are being designed, and will be installed during winter maintenance.
20013904

6/24-28
FG3-25
Not in correct position for TW elevation.
Computer system not yet installed, all diffusers at B-Branch are operated in manual
Operator corrected diffuser positioning.


6/24-25
FG6-12, FG6-13
Not in correct position for TW elevation.
Computer system not yet installed, all diffusers at Cascade Island are in manual
Operator corrected diffuser positioning


6/25-30
PH1 Collection channel velocity
Velocity meter registers less than 1.5 fps
Unknown
All other areas of the ladder were in criteria.  No action taken.


6/26
FG3-26
Not in correct position for TW elevation.
Computer system not yet installed, all diffusers at B-Branch are operated in manual
Operator corrected diffuser positioning.


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.

Date
Wash. Shore
Bradford Is.

6/24
17.5
17.3

6/25
17.3
17.2

6/26
17.3
17.2

6/27
17.4
17.3

6/28
17.5
17.4

6/29
17.8
17.1

6/30
18.1
17.9

4.4 Sensor Calibration.  Water level sensors were checked for calibration on 28 June this reporting week.  

4.5 Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.

5. MISCELLANEOUS

5.1 Fish were seen in the PH2 AWS triangle area again this year.  Mid-season dives showed the wall screen was intact.  Underwater video inspection proved the floor gratings were intact.  We suspect the fish get in through the wall screen as the spacing between the bars is greater that one inch.  To release fish trapped behind the screen, and to prevent more fish from becoming trapped, we pulled the two end screens up and dogged them off.  This action was coordinated through the district biologists.  We plan to cover the screens with expanded wire this winter maintenance season.

5.2 A contractor accidentally opened DSM2 ERG on the morning of 26 June.  It was immediately closed and the juvenile bypass system returned to normal without incident.  Flume flushing water moved fish in the transport pipe safely to the JMF.  PSMFC reported nothing unusual following the incident.

5.3 PH2 JBS injury/mortality issue: Juvenile mortality remains low at the JMF, according to PSMFC reports.  We plan no further action at this time.

5.4 The PH1 ice and trash sluiceway was closed from 1225 until 1405 on 28 June while USGS was in the forebay testing antenna range.

5.5 Spill stopped at 0000 16 June.  All bays are on seal during evening hours.  Bays 1 and 18 will remain open 4” during daylight hours.

                                                                                                                DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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